BU Open Access Articles

BU Open Access Articles

2018-06-01

Improving spatial prediction of Schistosoma haematobium prevalence in southern Ghana through new remote sensors and local water access profiles

Kulinkina AV, Walz Y, Koch M, Biritwum N-K, Utzinger J, Naumova EN (2018) Improving spatial prediction of Schistosoma haematobium prevalence in southern Ghana through new remote sensors and local water access profiles. PLoS Negl Trop Dis 12(6): e0006517. https://doi.org/10.1371/journal.pntd.0006 https://hdl.handle.net/2144/31495 Downloaded from DSpace Repository, DSpace Institution's institutional repository



Citation: Kulinkina AV, Walz Y, Koch M, Biritwum N-K, Utzinger J, Naumova EN (2018) Improving spatial prediction of *Schistosoma haematobium* prevalence in southern Ghana through new remote sensors and local water access profiles. PLoS Negl Trop Dis 12(6): e0006517. https://doi.org/10.1371/journal.pntd.0006517

Editor: Charles H. King, Case Western Reserve University School of Medicine, UNITED STATES

Received: September 16, 2017

Accepted: May 10, 2018

Published: June 4, 2018

Copyright: © 2018 Kulinkina et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Schistosomiasis prevalence data used in this study are reported in Supporting Information.

Funding: This study was funded in part by the National Institutes of Health (R34 Al097083-01A1), Tufts Institute for Innovation, Jonathan M. Tisch College of Civic Life, Natalie V. Zucker, Charlton, and Tufts Innovates grants. The funders had no role in study design, data collection and analysis,

RESEARCH ARTICLE

Improving spatial prediction of *Schistosoma haematobium* prevalence in southern Ghana through new remote sensors and local water access profiles

Alexandra V. Kulinkina¹*, Yvonne Walz², Magaly Koch^{1,3}, Nana-Kwadwo Biritwum⁴, Jürg Utzinger^{5,6}, Elena N. Naumova^{1,7}

1 Department of Civil and Environmental Engineering, Tufts University, Medford, United States of America, 2 Institute for Environment and Human Security, United Nations University, Bonn, Germany, 3 Center for Remote Sensing, Boston University, Boston, United States of America, 4 Neglected Tropical Diseases Program, Ghana Health Service, Accra, Ghana, 5 Swiss Tropical and Public Health Institute, Basel, Switzerland, 6 University of Basel, Basel, Switzerland, 7 Friedman School of Nutrition Science and Policy, Tufts University, Boston, United States of America

* alexandra.kulinkina@tufts.edu

Abstract

Background

Schistosomiasis is a water-related neglected tropical disease. In many endemic low- and middle-income countries, insufficient surveillance and reporting lead to poor characterization of the demographic and geographic distribution of schistosomiasis cases. Hence, modeling is relied upon to predict areas of high transmission and to inform control strategies. We hypothesized that utilizing remotely sensed (RS) environmental data in combination with water, sanitation, and hygiene (WASH) variables could improve on the current predictive modeling approaches.

Methodology

Schistosoma haematobium prevalence data, collected from 73 rural Ghanaian schools, were used in a random forest model to investigate the predictive capacity of 15 environmental variables derived from RS data (Landsat 8, Sentinel-2, and Global Digital Elevation Model) with fine spatial resolution (10–30 m). Five methods of variable extraction were tested to determine the spatial linkage between school-based prevalence and the environmental conditions of potential transmission sites, including applying the models to known human water contact locations. Lastly, measures of local water access and groundwater quality were incorporated into RS-based models to assess the relative importance of environmental and WASH variables.

Principal findings

Predictive models based on environmental characterization of specific locations where people contact surface water bodies offered some improvement as compared to the traditional decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

approach based on environmental characterization of locations where prevalence is measured. A water index (MNDWI) and topographic variables (elevation and slope) were important environmental risk factors, while overall, groundwater iron concentration predominated in the combined model that included WASH variables.

Conclusions/Significance

The study helps to understand localized drivers of schistosomiasis transmission. Specifically, unsatisfactory water quality in boreholes perpetuates reliance of surface water bodies, indirectly increasing schistosomiasis risk and resulting in rapid reinfection (up to 40% prevalence six months following preventive chemotherapy). Considering WASH-related risk factors in schistosomiasis prediction can help shift the focus of control strategies from treating symptoms to reducing exposure.

Author summary

Schistosomiasis is a water-related neglected tropical disease that disproportionately affects school-aged children in poor communities of low- and middle-income countries. Schistosomiasis transmission risk is affected by environmental, socioeconomic, and behavioral factors, including water, sanitation, and hygiene (WASH) conditions. We used fine spatial resolution (10-30 m) remotely sensed data, in combination with measures of local water access and groundwater quality, to predict schistosomiasis risk in 73 rural Ghanaian communities. We found that applying environmental models to specific locations where people contact surface water bodies (i.e., potential transmission locations), rather than to locations where prevalence is measured, improved model performance. A remotely sensed water index and topographic variables (elevation and slope) were important environmental risk factors, while overall, groundwater iron concentration predominated. In the study area, unsatisfactory water quality in boreholes perpetuates reliance of surface water bodies, indirectly increasing schistosomiasis risk and resulting in rapid reinfection (up to 40% prevalence six months following deworming). Considering WASH-related risk factors in schistosomiasis prediction can help shift the focus of control strategies from treating symptoms to reducing exposure.

Introduction

Schistosomiasis is an important parasitic disease that affects more than 250 million people [1]. Expressed in years lived with disability (YLDs), the impact of schistosomiasis is comparable to that of malaria (2.9 versus 3.2 million YLDs) [2]. Schistosomiasis is a disease of poverty, with 97% of all infections and 85% of the global at-risk population concentrated in Africa [3]. Ghana has an estimated country-wide prevalence of 23.3%, with focal, or localized, prevalence levels >50% [4].

Schistosomiasis is caused by infection with the trematode parasite of the genus *Schistosoma* [5]. Of the three species that commonly infect humans (*S. haematobium*, *S. mansoni*, and *S. japonicum*), the former two are prevalent in Africa [6]. *S. haematobium* is the predominant species in Ghana [4] and is the focus of the present study. Schistosomiasis has a complex life cycle that involves the parasite, intermediate host snails, and definitive human host (and

sometimes animal reservoir hosts). Transmission occurs in fresh surface water bodies that are contaminated with human waste, provide favorable ecologic conditions for intermediate host snails (*Bulinus* species for *S. haematobium*), and sustain human water contact [6]. Human transmission occurs when parasite larvae (cercariae) penetrate intact skin during water-based activities and has historically been most common in rural areas with natural slow flowing streams, ponds, and lakes [3,6].

To develop and implement effective control strategies against schistosomiasis, accurate data on the geographic and demographic distribution of infections are necessary. Surveillance in endemic low- and middle-income countries is inhibited by limited health infrastructure and cases evading clinical detection due to lower parasite burden and lessened symptoms that result from preventive chemotherapy with the anthelmintic drug praziquantel. Passive heath facility-based surveillance and reporting systems are known to severely underestimate the number of infections [7,8]. For example, a total of ~25,000 schistosomiasis cases were reported into the Ghanaian District Health Information Management System (DHIMS) in 2010 (data received from GHS, 2016). If only ~5 million children \leq 15 years of age residing in rural areas (i.e., high-risk population) [9] are considered at the estimated 23.3% infection rate [4], ~1.15 million cases would be expected. The reported cases represent only 2.2% of this expected number. Some correction for underreporting can be accomplished by predictive modeling, aiming to complement data from surveillance systems and field-based prevalence surveys.

Many schistosomiasis predictive modeling studies have been published and reviewed [10,11]. Most studies utilized remote sensing (RS) and geographic information system (GIS) approaches at large spatial extents (i.e., national, regional or continental) [12–14], with fewer applications of these methods to sub-national mapping [15–17]. Because snail populations, cercarial densities, human water contact patterns, and subsequent schistosomiasis infections exhibit strong spatial heterogeneity [10,18,19], further investigation of localized transmission drivers at smaller spatial extents is needed [10,11]. Furthermore, most studies included relatively few RS environmental predictors, mainly normalized difference vegetation index (NDVI), land surface temperature (LST), and elevation, whereas many other vegetation- and moisture-related indices and topographic variables are available and should be considered [11,20,21].

Another important limitation is that most studies utilized point-prevalence data of human infections (outcome) typically measured at schools, whereas RS-based environmental data (predictors) pertain to water bodies that serve as snail habitats and potential transmission locations. Most models do not account for this spatial mismatch between exposure and outcome measures [11]. A recent study used a more ecologically relevant approach, in which RS variables were extracted from geographically delineated water bodies within a buffer radius around the point-prevalence location [22]. An even more promising approach would be to apply the models to the specific locations along water bodies where human water contacts occur.

Further complicating the modeling approach at small spatial extents are socioeconomic and behavioral factors, including water, sanitation, and hygiene (WASH) conditions, known to affect individual schistosomiasis risk [23-25]. These factors may have an even greater bearing on the focal nature of disease distribution than the environment [26,27], and should be considered as predictors. While the inclusion of socioeconomic status and metrics of clean water and sanitation access have been advocated [10,11], to our knowledge, WASH variables have not yet been explicitly incorporated into spatial schistosomiasis predictive models.

The goal of the present study was to build upon existing predictive modeling approaches using *S. haematobium* prevalence data from 73 rural communities in the Eastern region of Ghana. We utilized fine resolution RS data (Landsat 8 and Sentinel-2), expanded the number of predictors (15 environmental and four WASH-related variables), and explored alternatives

for addressing the spatial mismatch between exposure and outcome measures. In this study, primary innovations include the use of a new RS data source (Sentinel-2), incorporation of field-mapped surface water contact sites into the RS-based environmental modeling approach, and exploration of WASH variables as additional schistosomiasis risk factors.

Methods

Ethics statement

The study was approved by the Institutional Review Board (IRB) at Tufts University in Boston, United States of America (protocol #11688) and Noguchi Memorial Institute for Medical Research in Accra, Ghana (protocol #1133). Letters of approval were obtained from national and regional offices of Ghana Health Service (GHS) and Ghana Education Service (GES). Written informed consent was obtained from the acting head teacher of each school that participated in the schistosomiasis prevalence survey. Verbal assent was sought from the participating children, an accepted ethical and practical approach used in similar low-risk studies [28].

Study area

The study was conducted in the tropical Eastern region (Fig 1), characterized by major and minor peak rainfall periods in June and October, respectively, with dry season lasting from November to February. Four major perennial rivers (Pra, Birim, Ayensu, and Densu) drain the region, with an abundance of smaller streams and ponds. Most of these water bodies are used extensively for domestic and recreational purposes (e.g., fetching, washing, swimming, and fishing). The Pra and Birim rivers, and some of their tributaries, however, are heavily polluted by alluvial gold mining and are no longer used due to high turbidity and presence of toxic compounds [29]. The region is relatively flat with some hilly areas and low mountains (Atiwa Mountain Range) reaching an elevation of approximately 750 m above sea level. The study area, spanning 10 administrative districts, was purposely selected outside of a 20-km buffer radius of Lake Volta [29]. Communities situated on its shores are historically known to be endemic for schistosomiasis [30]. However, little information is available about pockets of high transmission along minor rivers and streams that are not easily detected with RS technologies.

Community as a unit of analysis

Prior modeling studies mainly used point-prevalence as outcome data. Prevalence of *S. haema-tobium* eggs in urine samples (or hematuria as a proxy of infection) is typically measured at schools, while transmission may occur within some distance of this point-prevalence location. With extensive local knowledge from prior community-based studies [29,31–33], the present analysis was conducted at the "community" level. The spatial boundaries of communities were defined by Open Street Map (OSM) polygons (Fig 2) abstracted using QGIS software (version 2.12.3), an approach validated in a case study [29]. Subsequently, a buffer radius of 1 km was applied to each polygon. The buffer distance was chosen because nearly all known contact with water bodies occurred within 1 km of community boundaries. Throughout the manuscript, the term "community" refers to the OSM polygon + 1 km buffer area (Fig 2) and is used as a unit of analysis, also referred to as grain or support [21,34].

Data sources

Data for this study were obtained primarily from satellite RS sources and field studies, with some additional geographic features digitized from satellite imagery. Surface reflectance,



Fig 1. Map of the study area and spatial distribution of microhematuria (typical symptom of *S. haematobium* in school-aged children) prevalence created using the following data sources: Major rivers and town locations were obtained from CERSGIS, Accra, Ghana; hillshade relief surface was created from elevation data obtained from ASTER Global Digital Elevation Model (v2); mining locations were digitized from Sentinel-2 satellite imagery; microhematuria prevalence data were collected by A. Kulinkina.

https://doi.org/10.1371/journal.pntd.0006517.g001

thermal, and elevation data were obtained from RS sources. From these, vegetation and water indices, LST, and topographic variables were derived. WASH variables were obtained from field data available from past studies, namely global positioning system (GPS) coordinates of public water sources [29] and data about groundwater quality [32]. The outcome variable, *S. haematobium* prevalence (%) was measured in one school in each of the 73 study communities. Measures of improved and unimproved [35] water access and groundwater quality (WASH variables) were combined with RS-based variables to predict schistosomiasis prevalence across the study area. Data processing and analysis steps are described below and outlined in S1 and S2 Figs in Supporting Information.

Remotely sensed data. Surface reflectance data were obtained from two RS data sources: Landsat 8 Operational Land Imager (OLI) and Sentinel-2. Landsat 8 data were obtained from USGS Earth Explorer (http://earthexplorer.usgs.gov/) and included two cloud-free scenes that were mosaicked to cover the extent of the study area (Table 1). OLI data (bands 2–6 in Table 1) were downloaded as raw digital number (DN) values with a spatial resolution of 30 m and radiometrically and atmospherically corrected to obtain surface reflectance. This twostep procedure consisted of converting DN values to top-of-atmosphere (TOA) radiance, followed by an atmospheric correction using the Fast Line-of-sight Atmospheric Analysis of



Fig 2. Spatial definitions associated with the analysis conducted at the "community" level.

https://doi.org/10.1371/journal.pntd.0006517.g002

Hypercubes (FLAASH) module in ENVI 5.4 (Exelis Visual Information Solutions, Boulder, United States of America). Thermal data (bands 10 and 11 in Table 1) were downloaded from Landsat 8 Thermal InfraRed Sensor (TIRS) as level 1 (L1B) products with a spatial resolution of 100 m.

	Landsat 8	Sentinel-2
Scenes (acquisition	Path 193 Row 56 (December 22, 2015)	T30NXM (December 24, 2015)
dates)	Path 194 Row 56 (December 29, 2015)	T30NXN (December 24, 2015)
		T30NYM (December 24, 2015)
		T30NYN (December 24, 2015)
Bands (wavelengths)	Blue–B2 (0.450–0.515 μm)	Blue–B2 (0.490 μm)
	Green-B3 (0.525-0.600 μm)	Green–B3 (0.560 μm)
	Red-B4 (0.630-0.680 μm)	Red–B4 (0.665 μm)
	Near infrared–B5 (0.845–0.885 μm)	Near infrared–B8 (0.842 μm)
	Short wavelength infrared–B6 (1.560–	Short wavelength infrared-B11
	1.660 μm)	(1.610 µm)
	Long wavelength infrared-B10 (10.30-	
	11.30 μm)	
	Long wavelength infrared-B11 (11.50-	
	12.50 μm)	

Table 1. Summary of surface reflectance data used in the study.

https://doi.org/10.1371/journal.pntd.0006517.t001

Sentinel-2 surface reflectance data (bands 2, 3, 4, 8, and 11 in <u>Table 1</u>) were obtained from Copernicus Data Hub (https://scihub.copernicus.eu/). Four cloud-free scenes were mosaicked to cover the extent of the study area (<u>Table 1</u>). TOA radiance (level 1C) products were downloaded with a spatial resolution of 10 or 20 m and converted to level 2A surface reflectance by applying the atmospheric correction using the Sen2Cor processor in open-source Sentinel Application Platform (SNAP) software (version 5.0).

ASTER Global Digital Elevation Model (GDEM v2) data were obtained from USGS Global Data Explorer (gdex.cr.usgs.gov) with a spatial resolution of 30 m. A moving window (3x3) majority filter was applied to the elevation data to eliminate image artefacts [36,37] using the Spatial Analyst extension in ArcGIS 10.2.2.

Settlement data were obtained from the German Aerospace Center (http://www.dlr.de) as a new Global Urban Footprint (GUF) product. GUF is a binary raster data product of populated and unpopulated pixels produced from 2011–2012 TerraSAR-X and TanDEM-X radar images [38]. GUF was chosen as a source of settlement data due to its 0.4 arcsec geometric resolution, or 12 m spatial resolution, which most closely matched the resolution of the other spatial data used in the study.

Field data. A cross-sectional *S*. *haematobium* prevalence survey was conducted in May and June 2016 in the largest primary school in each of the 73 study communities (population range 500–5,000). The most recent round of national school-based preventive chemotherapy had been conducted in January 2016 (six months prior to the survey); all study schools had participated, with an average treatment coverage of 78% (data provided by GHS, 2016). All children in grades 3 and 4 (age range 8–13 years) who expressed verbal assent were enrolled into the study. Upon detailed demonstrations of the specimen collection procedure, children were invited to provide a urine sample between 10:00 and 14:00 hours that was tested for microhematuria using a semi-quantitative reagent strip on-site. Samples with any blood presence, including "trace", were categorized as positive readings [28]. Infected children were offered praziquantel according to their weight by a local nurse or community health worker in a private location. No identifying information about study subjects was recorded besides school/community name, sex, and grade.

A total of 5,220 children (2,802 boys and 2,418 girls) were registered in grades 3 and 4 in the 73 study schools. Of these, 3,746 children (72%) were present on the day of screening. Attendance in some of the schools was as low as 46%. A total of 3,628 children (97%) were enrolled into the study, and 3,612 (>99%) provided urine samples for analysis. Prevalence of microhematuria in the study population was 14%; school-level prevalence values ranged between 0 and 40% (Fig I; S1 Table, Supporting Information).

Data processing

Six environmental indices were calculated from Landsat 8 (OLI) and Sentinel-2 surface reflectance data (Table 2) in R software (version 3.3.1). In the enhanced vegetation index (EVI) equation, *L* value adjusts for canopy background and *C* values are coefficients for atmospheric resistance. These enhancements allow for index calculation as a ratio between the red and the near infrared (*nir*) band values, while reducing the background and atmospheric noise and saturation [39]. The values of $C_1 = 6$, $C_2 = 7.5$, and L = 1 were obtained from the Landsat 8 product guide [40]. In the soil adjusted vegetation index (SAVI) equation, *L* is the soil calibration factor that minimizes soil background conditions that affect partial canopy spectra. The *L* value of 0.5 minimizes soil brightness variation and eliminates the need for additional calibration for different soils [41]. Landsat 8 (TIRS) thermal data were processed using ATCOR [42]

Index	Equation	Reference
Normalized difference vegetation index (NDVI)	$\frac{(nir-red)}{(nir+red)}$	[43]
Enhanced vegetation index (EVI)	$\frac{(nir-red)}{(nir+C_1*red-C_2*blue+L)}$	[39]
Soil adjusted vegetation index (SAVI)	$\frac{(nir-red)}{(nir+red+L)}(1+L)$	[41]
Modified soil adjusted vegetation index (MSAVI)	$\frac{\left[2*nir+1-\sqrt{\left(2*nir+1\right)^2-8(nir-red)\right]}}{2}$	[44]
Normalized difference water index (NDWI)	(green-nir) (green+nir)	[45]
Modified normalized difference water index (MNDWI)	(green-swir) (green+swir)	[46]

https://doi.org/10.1371/journal.pntd.0006517.t002

with a standard emissivity of 0.985 to detect water surface temperature, and converted from Kelvin (K) to degrees Celsius (°C) to represent LST.

Elevation data were used to derive stream order and slope. Topographic drainage lines were delineated from the digital elevation model (DEM) based on the potential flow direction from higher to lower elevation and accumulation of surface runoff according to topographic conditions using Arc Hydro Tools in ArcGIS (version 10.2.2). The resulting stream network was ordered according to Strahler [47]. Slope of the terrain was derived from the DEM as a proxy indicator for potential flow velocity of surface runoff with inclination calculated in degrees.

GPS coordinates of public water sources (standpipes (SPs), boreholes (BHs), protected and unprotected hand-dug wells (HDWs), and surface water access points (SWAPs)) were available from a prior study [29]. SPs, BHs, and protected HDWs that were functional at the time of the study constituted functional improved water sources (FIWS) that are not capable of transmitting schistosomiasis. SWAPs constituted unimproved water sources that are capable of transmitting schistosomiasis. Two categorical raster layers were derived from the GPS data using a buffer analysis conducted in ArcGIS 10.2.2, which represented improved water access (within 100–500 m of FIWS) and surface water access (within 100–500 m of SWAP), to test the hypothesis that locations closer to FIWSs have a lower risk of schistosomiasis transmission and locations closer to SWAPs have higher risk of schistosomiasis transmission [29].

Two additional raster layers of interpolated groundwater iron and TDS concentrations (mg/l) were also obtained from a prior study [32]. Groundwater quality variables were included because prior studies [29,32,33] suggested that elevated iron and TDS concentrations in BHs may increase reliance on contaminated surface water bodies, thereby potentially serving as indirect risk factors for schistosomiasis transmission.

Lastly, *S. haematobium* prevalence (% positive samples) was calculated from survey data. Prevalence was determined separately for boys and girls in each grade and then adjusted to a gender- and grade- balanced population using direct standardization [48]. Standardized school-level point-prevalence values (S1 Table, Supporting Information) were taken to represent community-level prevalence based on the following validated [49] assumptions: (i) microhematuria prevalence measured by reagent strip is a reasonable proxy of *S. haematobium* prevalence in a presumably lightly infected population due to recent preventive chemotherapy; (ii) 3rd and 4th grade school children are a representative study population; and (iii) where a child lives and attends school are not spatially dependent, inferring that prevalence value at one school is representative of community-level prevalence.

Variable extraction and aggregation

A total of 15 environmental and four WASH predictor variables (<u>Table 3</u>; <u>S3–S21</u> Figs, Supporting Information) were derived and resampled to a matching spatial resolution of 10 m.

Data source	Variable	Scale [range]	Resolution [m]	Mask	Aggregation
OLI/Sentinel	Blue band	Continuous [0-1]	30 / 10	1, 2, 4, 5, 6	Median
OLI/Sentinel	Green band	Continuous [0-1]	30 / 10	1, 2, 4, 5, 6	Median
OLI/Sentinel	Red band	Continuous [0–1]	30 / 10	1, 2, 4, 5, 6	Median
OLI/Sentinel	nir band	Continuous [0-1]	30 / 10	1, 2, 4, 5, 6	Median
OLI/Sentinel	swir band	Continuous [0-1]	30 / 20	1, 2, 4, 5, 6	Median
OLI/Sentinel	NDVI	Continuous [-1-1]	30 / 10	1, 2, 4, 5, 6	Median
OLI/Sentinel	EVI	Continuous [-1-1]	30 / 10	1, 2, 4, 5, 6	Median
OLI/Sentinel	SAVI	Continuous [-1-1]	30 / 10	1, 2, 4, 5, 6	Median
OLI/Sentinel	MSAVI	Continuous [-1-1]	30 / 10	1, 2, 4, 5, 6	Median
OLI/Sentinel	NDWI	Continuous [-1-1]	30 / 10	1, 2, 4, 5, 6	Median
OLI/Sentinel	MNDWI	Continuous [-1-1]	30 / 10	1, 2, 4, 5, 6	Median
TIRS	LST [°C]	Continuous [13-47]	100	1, 2, 4, 5, 6	Median
DEM	Elevation [m]	Continuous [1-870]	30	1, 2, 4, 5, 6	Median
DEM	Slope [°]	Continuous [0-85]	30	1, 2, 4, 5, 6	Median
DEM	Stream order	Categorical [0-5]	30	1	Max
Field data	FIWS access	Categorical [0–5]		3	Mode
Field data	SWAP access	Categorical [0–5]		3	Mode
Field data	Iron [mg/l]	Continuous [0.1-0.7]		1	Median
Field data	TDS [mg/l]	Continuous [83–616]		1	Median

Table 3. Environmental (top) and WASH (bottom) predictor variables.

https://doi.org/10.1371/journal.pntd.0006517.t003

While *S. haematobium* infection prevalence was represented by point data, predictors were represented by continuous raster data (Fig 3). Therefore, extraction and aggregation of the raster data within the "community" polygons were necessary.

A total of six methods of variable extraction (masks) were used (Fig 3): none {1}-all pixels within the "community" polygon were extracted; unpopulated {2}-data were extracted only for unpopulated pixels as defined by the GUF data; populated {3}-data were extracted only for populated pixels as defined by the GUF data; all water bodies {4}-mask was derived by combining the topographic drainage lines from the DEM, supplemented with ponds, lakes, and gold mining pits that were digitized from satellite imagery; unmined water bodies {5}-mask was derived by removing water bodies that are known to be affected by mining from "all water bodies"; SWAPs {6}-defined as the single pixel GPS points of known surface water contact sites.

To understand the spatial linkage between school-based prevalence and the environmental conditions, almost all environmental variables were extracted using masks {1, 2, 4, 5, and 6} (Table 3), listed in the order of increasing ecologic relevance. For example, the most ecologically relevant method is to match school-based schistosomiasis prevalence with environmental variables extracted from points within the "community" where known contact with water bodies occurs (m6). Method 3 (populated areas) was not relevant for environmental variable extraction because these locations are not representative of schistosomiasis transmission. Conversely, measures of safe (FIWS) and unsafe (SWAP) water access apply only to populated areas; hence only method 3 was used to extract these two WASH variables. Unmasked data (m1) were used to extract stream order, iron, and TDS concentrations (Table 3). For aggregation of environmental variables, primarily the median pixel values were used, except for stream order, where maximum value was used. For aggregation of WASH variables, either median (iron and TDS) or mode (FIWS and SWAP access) were used (Table 3).



Six methods of variable extraction (masks)*



* Raster data are extracted for blue areas only

Fig 3. Modeling approach explaining raster data extraction methods to be matched to each point-prevalence location.

https://doi.org/10.1371/journal.pntd.0006517.g003

Data analysis

Exploratory analyses included variable summaries and correlations, followed by random forest models. The random forest approach was chosen because it can deal with continuous outcome data, multicollinear predictor variables, and low numbers of training samples, it is the recommended machine learning method for generating predictions [50], and it has been successfully applied in similar studies [22].

Five non-parametric random forest models were conducted with 15 environmental predictor variables (Table 3) to determine which of the five masks presented the best method of variable extraction. Two versions of the analyses were conducted in parallel (with Landsat 8 and Sentinel-2 surface reflectance values and environmental indices) to test consistency of predictive performance of RS data obtained from these two satellites with similar acquisition dates. Explanatory power of random forest models was compared using root-mean-square error (RMSE) and R² values [51], and relative importance of predictor variables was assessed using the increasing node purity ("IncNodePurity") metric [52,53].

All models were applied back to the raster stack of predictor variables to derive continuous predicted *S. haematobium* prevalence surfaces. Although predicted values were available for all

pixels, the same masks used to extract the explanatory variables were applied to the respective predicted prevalence surfaces. After applying the masks, the median predicted values within each "community" were plotted against observed prevalence values. The quality of prediction was assessed using Spearman's rank correlation between model predicted and observed values, and their fit was compared to the line of equality. Lastly, environmental data extracted using the best performing mask were combined with the WASH variables in a final model to assess the relative importance of the two groups of variables.

Results

Comparison of five environmental variable extraction methods

As an exploratory analysis, Spearman's rank correlations were computed between pairs of environmental indices (S2 Table, Supporting Information). The correlation values were consistent across extraction masks and across RS data sources. As expected, correlations among the vegetation indices derived using both Landsat 8 and Sentinel-2 data were generally very high (0.90–0.99). Lower correlation values were observed between the two water indices NDWI and MNDWI (~0.70). Consequently, negative correlation values between NDWI and the vegetation indices were much higher than those between MNDWI and the vegetation indices (0.91 versus 0.50).

To explore the potential reason for this, NDWI and MNDWI were visually compared against a map (Fig 4). In the first row (A1 and B1), schematic maps of study communities are shown with populated areas indicated in gray and water bodies, comprised of rivers/streams and dug mining pits, indicated in blue. It appears that the NDWI computed with Landsat 8 data (A2 and B2) results in false detection of water bodies (i.e., misclassification of developed surfaces such as settlements and roads), essentially serving as an inverse of a vegetation index, which explains the strong negative correlation with vegetation indices. On the other hand, the MNDWI (A3 and B3) more precisely detects water bodies, particularly mining pits. Same conclusions apply to NDWI and MNDWI values derived from Sentinel-2 data (S13 and S14 Figs, Supporting Information). Neither index performed adequately at detecting the SWAPs, shown as + symbols in Fig 4.

Random forest models were first run for each extraction method using environmental variables only (Table 4). Two versions of the environmental models were run in parallel with Landsat 8 and Sentinel-2 surface reflectance and environmental indices (in addition to LST and topographic variables derived from a single source). The R² values for all models were relatively low (<0.20), indicating that environmental variables alone were not able to describe more than 15–20% of the variability in *S. haematobium* prevalence, regardless of RS data source or extraction mask. The predicted prevalence at the pixel level ranged from approximately 5% to 28% (Fig.4). Aggregated predicted community-level prevalence ranged between 7% and 22%, as compared to the observed prevalence range of 0–40%.

Correlations between observed and predicted prevalence values were higher on average for models produced using Landsat 8 environmental data as compared to Sentinel-2 data (both in combination with LST and topographic variables). Models derived using the SWAP mask produced the highest correlation values using both Landsat 8 (r = 0.76, p < 0.01) and Sentinel-2 data (r = 0.67, p < 0.01) (Table 4). However, scatter plots of observed versus predicted values still deviated substantially from the line of equality (S22 and S23 Figs, Supporting Information) due to the overall low R² values. From a visual assessment of the predicted prevalence surfaces produced using environmental variables (Fig 5; S24–S33 Figs, Supporting Information), it appears that the SWAP mask resulted in more precise prediction, including correct delineation of water bodies as high-risk locations (Fig 5, panel A6).



Fig 4. Schematic images of study communities showing settlements and water bodies (A1 and B1), NDWI values (A2 and B2), and MNDWI values (A3 and B3) generated using Landsat 8 data.

https://doi.org/10.1371/journal.pntd.0006517.g004

Variable importance was also explored using the IncNodePurity measure from random forest models (S22 and S23 Figs, Supporting Information). MNDWI was an important water index, particularly when environmental data were extracted without knowledge of water contact sites (masks 1, 2, 4, and 5). Vegetation indices were not commonly observed among the

Table 4. Results of environmental models for various extraction masks showing the R² value and Spearman's rank correlation value (r) between model predicted and observed prevalence values.

Mask*	R ² (RMSE)	r (p-value)	R ² (RMSE)	r (p-value)
	Landsat 8 data	Sentinel-2 data		
None {1}	0.14 (9.36)	0.49 (< 0.01)	0.14 (9.65)	0.33 (< 0.01)
Unpopulated {2}	0.17 (9.07)	0.46 (< 0.01)	0.14 (9.80)	0.28 (< 0.01)
All WBs {4}	0.15 (9.62)	0.48 (< 0.01)	0.14 (9.70)	0.40 (< 0.01)
Unmined WBs {5}	0.12 (9.68)	0.51 (< 0.01)	0.12 (9.71)	0.34 (< 0.01)
SWAPs {6}	0.15 (9.47)	0.76 (< 0.01)	0.13 (9.43)	0.67 (< 0.01)

* The number in brackets refers to the mask in Fig 3

https://doi.org/10.1371/journal.pntd.0006517.t004

PLOS | NEGLECTED TROPICAL DISEASES



Fig 5. Predicted prevalence from Landsat 8 data (A) and Sentinel-2 data (B) for five extraction masks {1, 2, 4, 5, and 6}. Surface water access points are shown as + symbols. The scale and extent of the image match Fig 4(B).

https://doi.org/10.1371/journal.pntd.0006517.g005





Fig 6. Predicted prevalence for the entire study area; for two smaller zoom windows (A and B); and variable importance values for the final model conducted with Landsat 8 environmental, topographic, and WASH variables. The scales and extents of A and B match Figs 4 and 5. Surface water access points are shown as + symbols.

https://doi.org/10.1371/journal.pntd.0006517.g006

top five important variables in the Landsat 8 models; EVI and NDVI were the most important vegetation indices in the Sentinel-2 models. Slope and elevation were important in many models, whereas stream order was always the least important variable.

Contribution of WASH variables

The final model consisted of environmental variables derived from Landsat 8 data using the SWAP mask in combination with WASH variables. The addition of WASH variables only slightly increased the R² value from 0.15 to 0.17 and decreased the RMSE from 9.47 to 9.03. However, iron concentration became by far the most important variable. The importance of iron concentration was also evident in the predicted prevalence surfaces, with high values on the western side of the Atiwa Mountain Range (Fig 6) coinciding with high groundwater iron content (S21 Fig, Supporting Information). FIWS and SWAP access indicators were not important in the final model. Of the environmental variables, elevation remained important and stream order remained unimportant (Fig 6). The correlations between predicted and observed values were not extracted for the final models because multiple masks were used in the model.

Discussion

In this study, we utilized publicly available environmental data from two multispectral optical sensors in combination with topographic variables and field-collected WASH variables to

assess their performance in predicting *S. haematobium* prevalence at a sub-national spatial extent. Furthermore, we tested five methods of environmental data extraction with varying degrees of ecologic relevance. In epidemiologic literature, schistosomiasis is known as a focal disease, meaning that neighboring villages with seemingly similar conditions can have drastically different transmission profiles and disease prevalence levels [10,18,19]. This study attempted to characterize some of the sources of spatial heterogeneity at small spatial extents using fine resolution RS data and WASH-related risk factors.

We found that knowledge of water contact sites shows promise in schistosomiasis risk prediction at small spatial extents. According to a visual assessment, environmental data extracted using the SWAP mask more precisely delineated water bodies as high-risk locations within communities (Fig 5). This mask also produced the highest correlation between model predicted and observed prevalence values, depicting heterogeneity in transmission risk among communities (Table 4).

Of the two water indices we explored, MNDWI was the preferred index due to more accurate detection of water bodies. NDWI values were equally high for water and developed pixels (roads and settled areas), indicating false detection of water bodies. Generally, higher values of MNDWI correlated with higher schistosomiasis risk. However, even MNDWI could not detect small streams that sustained most of surface water use (i.e., SWAPs). Further investigation of these two indices and their utility in water-related disease modeling is recommended. Vegetation indices did not play a major role in prediction. This is not surprising, especially in the SWAP mask models, as these indices are likely characterizing land vegetation cover, rather than aquatic vegetation that affects intermediate host snail abundance [11].

LST did not exhibit a strong influence on schistosomiasis risk, most probably due to the lack of variability in LST values (25–32 °C), all of which were well within the favorable temperature range for snail and cercariae survival [54,55]. Furthermore, because the water bodies in the study area are very small, the spatial resolution of the temperature data (100 m) was likely too coarse to detect water temperature.

Slope and elevation were important in prediction. Higher elevation correlated with higher schistosomiasis risk, counter to the literature, likely because the Atiwa Mountain Range is still quite low in elevation, far below the 2,000-m above sea level threshold for *S. haematobium* transmission [18]. Higher slope correlated with lower schistosomiasis risk, potentially due to faster stream flows. At water velocities > 0.3 m/s, snails can become dislodged and swept away [55]. Surprisingly, stream order was consistently the least important variable in all models, while it demonstrated a significant positive association with schistosomiasis risk in other studies [17,22]. A potential explanation for this is the abundance of small streams throughout the study communities, widespread preference of people for surface water over groundwater, and hence their uniform extensive use.

In our study, variables of improved and unimproved water access were not predictive of schistosomiasis risk, consistently with the findings of Lai et al. [4]. However, high iron concentration in groundwater was associated with increased schistosomiasis risk. Our prior studies have provided qualitative support for the hypothesis that unfavorable groundwater quality in improved water sources (i.e., boreholes and piped water systems) for drinking and laundry is a significant driver of increased surface water use, serving as an indirect risk factor for schistosomiasis transmission. The final model results confirmed this hypothesis, with groundwater iron content being the predominant schistosomiasis risk factor with a much higher IncNodePurity value as compared to any of the environmental variables (Fig 6). Indeed, in Fig 6, the area with high predicted schistosomiasis prevalence area in the center of the image corresponds to the high iron concentration cluster (S21 Fig, Supporting Information).

Overall, the models had relatively low predictive power and predicted prevalence values deviated substantially from the observed values, indicating overprediction in the low-prevalence range and underprediction in the high-prevalence range. This is most likely due to the effects of preventive chemotherapy on prevalence measures. With increased treatment frequency, it becomes difficult to detect the effects of environmental conditions on transmission risk [4,22]. It would be valuable to apply these approaches in similar geographic extents with a wider prevalence range. Exploring different methods of defining "communities" over which risk factor variables are aggregated (e.g., varying the buffer radius within which transmission occurs) in other geographic, demographic, and cultural contexts is also recommended.

We also found that Landsat 8 and Sentinel-2 sensors with similar radiometric resolutions (12-bit) and acquisition dates (all images were acquired within one week), on average, had similar predictive capacities. Cloud cover presented a substantial challenge in RS data acquisition from both data sources, with few cloud-free images available only in the dry season (December and January). Additionally, Landsat 8 data were more affected by haze and ocean spray, as compared to Sentinel-2 data. As RS data algorithms improve, future studies should consider repeating the same environmental models using RS data representative of both dry and rainy seasons to analyze the impact of water stability and dynamics. Synthetic Aperture Radar (SAR) data (e.g., from Sentinel-1A) could provide additional information in this and similar cloud-affected regions.

Apart from technical challenges associated with using RS data, several logistic challenges may have affected the quality of this study. First, low attendance in some of the study schools (range 46–95%) associated with sporting events and market days may have affected the prevalence measures. For example, children from agrarian families who were absent on market days are likely different in terms of socioeconomic status and schistosomiasis exposure profile from those who were present and participated in the study. In a smaller study, it would have been possible to go back and screen absentees; in the present study, this was not possible due to time and scheduling limitations and absence of identifying information about participants. Additional challenges arose from working across 10 administrative districts, especially with securing local GHS personnel to administer praziquantel. Scheduling and coordination efforts were further complicated by the community health workers being on strike in some of the districts during the study.

Despite the challenges and limitations, our study makes important contributions to the modeling approaches of schistosomiasis transmission at small spatial extents. First, knowledge of human water contact sites bridges the gap between where prevalence is measured and where transmission may have occurred. This is a critical gap in models that utilize environmental data as predictors of human infection. Second, the impact of groundwater iron concentrations on schistosomiasis risk. With prevalence rates up to 40% only six months after preventive chemotherapy and very high rates of fetching surface water (up to 100%) and swimming (up to 90%) [49], reinfection is a major concern in the study area. Groundwater quality in improved water sources, more so than improved water access in general, plays a major role in reinfection patterns and can impede schistosomiasis control. While it is well-established that preventive chemotherapy reduces prevalence and worm burden in the short term, with rapid reinfection, it cannot have more than a temporary effect on transmission without complementary improvements in WASH [23,24,56]. Our extensive experience in the Eastern region of Ghana suggests that it is not only increasing access to WASH resources that matters, but rather increasing utilization of these resources in accordance with local perceptions and preferences. Considering WASH-related risk factors in schistosomiasis prediction can help shift the focus of control strategies from treating symptoms to reducing exposure [56].

Supporting information

S1 Table. Microhematuria prevalence survey results. (XLSX)

S2 Table. Spearman's rank correlations among six environmental indices (all are statistically significant; p < 0.05); Sentinel-2 values are shown in top and Landsat 8 in bottom of the matrix.

(XLSX)

S1 Fig. Data processing steps. (TIF)

S2 Fig. Data analysis steps. (TIF)

S3 Fig. Sentinel-2 blue band reflectance values. (TIF)

S4 Fig. Sentinel-2 green band reflectance values. (TIF)

S5 Fig. Sentinel-2 red band reflectance values. (TIF)

S6 Fig. Sentinel-2 near infrared band reflectance values. (TIF)

S7 Fig. Sentinel-2 short wavelength infrared band reflectance values. (TIF)

S8 Fig. Landsat 8 land surface temperature values. (TIF)

S9 Fig. Sentinel-2 NDVI values. (TIF)

S10 Fig. Sentinel-2 EVI values. (TIF)

S11 Fig. Sentinel-2 SAVI values. (TIF)

S12 Fig. Sentinel-2 MSAVI values. (TIF)

S13 Fig. Sentinel-2 NDWI values. (TIF)

S14 Fig. Sentinel-2 MNDWI values. (TIF)

S15 Fig. Elevation values (in meters) derived from GDEM. (TIF)

S16 Fig. Slope values (in degrees) derived from GDEM. (TIF)

S17 Fig. Slope values derived from GDEM. (TIF)

S18 Fig. Functional improved water source (FIWS) access values derived in ArcGIS from field data.

(TIF)

S19 Fig. Perennial surface water source (SWAP) access values derived in ArcGIS from field data.

(TIF)

S20 Fig. Interpolated total dissolved solids (TDS) concentration values (in mg/L) derived in ArcGIS from field data.

(TIF)

S21 Fig. Interpolated iron concentration values (in mg/l) derived in ArcGIS from field data. (TIF)

S22 Fig. Scatter plots of model predicted (x-axis) vs. observed (y-axis) prevalence values as compared to the line of equality [left]; variable importance values [right] for random forest models conducted with environmental variables from Landsat 8 and topographic variables from GDEM.

(TIF)

S23 Fig. Scatter plots of model predicted (x-axis) vs. observed (y-axis) prevalence values as compared to the line of equality [left]; variable importance values [right] for random forest models conducted with environmental variables from Sentinel-2 and topographic variables from GDEM.

(TIF)

S24 Fig. Predicted prevalence values using Landsat 8 data (mask 1). (TIF)

S25 Fig. Predicted prevalence values using Sentinel-2 data (mask 1). (TIF)

S26 Fig. Predicted prevalence values using Landsat 8 data (mask 2). (TIF)

S27 Fig. Predicted prevalence values using Sentinel-2 data (mask 2). (TIF)

S28 Fig. Predicted prevalence values using Landsat 8 data (mask 4). (TIF)

S29 Fig. Predicted prevalence values using Sentinel-2 data (mask 4). (TIF)

S30 Fig. Predicted prevalence values using Landsat 8 data (mask 5). (TIF)

S31 Fig. Predicted prevalence values using Sentinel-2 data (mask 5). (TIF)

S32 Fig. Predicted prevalence values using Landsat 8 data (mask 6). (TIF)

S33 Fig. Predicted prevalence values using Sentinel-2 data (mask 6). (TIF)

Acknowledgments

We wish to acknowledge Ghana Health Service and Ghana Education Service for approval and support of the study. We thank Michael N. Adjei, Bernard O. Gyamfi, Rachel Martel, and Liliana Schmitt for assistance with data collection, and Zita Sebesvari for commenting on the manuscript. We are also thankful to members of the traditional leadership of the study communities and school administrators for allowing us to conduct the study in their communities, and to school children for their participation.

Author Contributions

Conceptualization: Alexandra V. Kulinkina, Yvonne Walz, Elena N. Naumova.

Data curation: Alexandra V. Kulinkina.

Formal analysis: Alexandra V. Kulinkina, Yvonne Walz, Magaly Koch, Elena N. Naumova.

Funding acquisition: Alexandra V. Kulinkina, Elena N. Naumova.

Investigation: Alexandra V. Kulinkina, Nana-Kwadwo Biritwum, Elena N. Naumova.

Methodology: Alexandra V. Kulinkina, Yvonne Walz.

Project administration: Alexandra V. Kulinkina.

Resources: Alexandra V. Kulinkina.

Software: Alexandra V. Kulinkina, Magaly Koch.

Supervision: Alexandra V. Kulinkina, Nana-Kwadwo Biritwum, Elena N. Naumova.

Validation: Alexandra V. Kulinkina, Jürg Utzinger.

Visualization: Alexandra V. Kulinkina.

Writing – original draft: Alexandra V. Kulinkina.

Writing – review & editing: Alexandra V. Kulinkina, Yvonne Walz, Magaly Koch, Nana-Kwadwo Biritwum, Jürg Utzinger, Elena N. Naumova.

References

- Hotez PJ, Alvarado M, Basáñez MG, Bolliger I, Bourne R, Boussinesq M, et al. (2014). The Global Burden of Disease Study 2010: interpretation and implications for the neglected tropical diseases. PLoS Negl Trop Dis 8: e2865. https://doi.org/10.1371/journal.pntd.0002865 PMID: 25058013
- Vos T, Barber RM, Bell B, Bertozzi-Villa A, Biryukov S, Bolliger I, et al. (2015). Global, regional, and national incidence, prevalence, and years lived with disability for 301 acute and chronic diseases and injuries in 188 countries, 1990–2013: a systematic analysis for the Global Burden of Disease Study 2013. Lancet 386: 743–800. https://doi.org/10.1016/S0140-6736(15)60692-4 PMID: 26063472
- Steinmann P, Keiser J, Bos R, Tanner M, Utzinger J (2006). Schistosomiasis and water resources development: systematic review, meta-analysis, and estimates of people at risk. Lancet Infect Dis 6: 411–425. https://doi.org/10.1016/S1473-3099(06)70521-7 PMID: 16790382
- Lai Y, Biedermann P, Ekpo UF, Garba A, Mathieu E, Midzi N, et al. (2015). Spatial distribution of schistosomiasis and treatment needs in sub-Saharan Africa: a systematic review and geostatistical analysis. Lancet Infect Dis 15: 927–940. https://doi.org/10.1016/S1473-3099(15)00066-3 PMID: 26004859
- Colley DG, Bustinduy AL, Secor WE, King CH (2014). Human schistosomiasis. Lancet 383: 2253– 2264. https://doi.org/10.1016/S0140-6736(13)61949-2 PMID: 24698483

- 6. Gryseels B, Polman K, Clerinx J, Kestens L (2006). Human schistosomiasis. Lancet 368: 1106–1118. https://doi.org/10.1016/S0140-6736(06)69440-3 PMID: 16997665
- Liang S, Yang C, Zhong B, Guo J, Li H, Carlton EJ, et al. (2014). Surveillance systems for neglected tropical diseases: global lessons from China's evolving schistosomiasis reporting systems, 1949–2014. Emerg Themes Epidemiol 11: 19. https://doi.org/10.1186/1742-7622-11-19 PMID: 26265928
- Wrable M, Kulinkina AV, Liss A, Koch M, Cruz M, Biritwum NK, et al. (2017). The use of remotely sensed environmental parameters for schistosomiasis prediction across climate zones in Ghana. Environ Monit Assess (in press).
- Ghana Statistical Service. 2010 Population & Housing Census [Internet] 2013 [cited 2018 Apr 05]. http://www.statsghana.gov.gh/docfiles/publications/2010 PHC_National_Analytical_Report.pdf
- Simoonga C, Utzinger J, Brooker S, Vounatsou P, Appleton CC, Stensgaard AS, et al. (2009). Remote sensing, geographical information system and spatial analysis for schistosomiasis epidemiology and ecology in Africa. Parasitology 136: 1683–1693. https://doi.org/10.1017/S0031182009006222 PMID: 19627627
- Walz Y, Wegmann M, Dech S, Raso G, Utzinger J (2015). Risk profiling of schistosomiasis using remote sensing: approaches, challenges and outlook. Parasit Vectors 8: 163. <u>https://doi.org/10.1186/</u> s13071-015-0732-6 PMID: 25890278
- 12. Ekpo UF, Hürlimann E, Schur N, Oluwole AS, Abe EM, Mafe MA, et al. (2013). Mapping and prediction of schistosomiasis in Nigeria using compiled survey data and Bayesian geospatial modelling. Geospat Health 7: 355–366. https://doi.org/10.4081/gh.2013.92 PMID: 23733296
- Soares Magalhães RJ, Biritwum NK, Gyapong JO, Brooker S, Zhang Y, Blair L, et al. (2011). Mapping helminth co-infection and co-intensity: geostatistical prediction in Ghana. PLoS Negl Trop Dis 5: e1200. https://doi.org/10.1371/journal.pntd.0001200 PMID: 21666800
- Schur N, Hürlimann E, Garba A, Traore MS, Ndir O, Ratard RC, et al. (2011). Geostatistical modelbased estimates of schistosomiasis prevalence among individuals aged ≤20 years in West Africa. PLoS Negl Trop Dis 5: e1194. https://doi.org/10.1371/journal.pntd.0001194 PMID: 21695107
- Walz Y, Wegmann M, Dech S, Vounatsou P, Poda JN, N'Goran EK, et al. (2015). Modeling and validation of environmental suitability for schistosomiasis transmission using remote sensing. PLoS Negl Trop Dis 9: e0004217. https://doi.org/10.1371/journal.pntd.0004217 PMID: 26587839
- Brooker S, Hay SI, Issae W, Hall A, Kihamia CM, Lwambo NJS, et al. (2001). Predicting the distribution of urinary schistosomiasis in Tanzania using satellite sensor data. Trop Med Int Heal 6: 998–1007.
- Beck-Wörner C, Raso G, Vounatsou P, N'Goran EK, Rigo G, Parlow E, et al. (2007). Bayesian spatial risk prediction of *Schistosoma mansoni* infection in western Côte d'Ivoire using a remotely-sensed digital elevation model. Am J Trop Med Hyg 76: 956–963. PMID: 17488922
- Brooker S, Michael E (2000). The potential of geographical information systems and remote sensing in the epidemiology and control of human helminth infections. Adv Parasitol 47: 245–288. PMID: 10997209
- 19. Brooker S, Hay SI, Bundy DAP (2002). Tools from ecology: useful for evaluating infection risk models? Trends Parasitol 18: 70–74. PMID: 11832297
- 20. Herbreteau V, Salem G, Souris M, Hugot JP, Gonzalez JP (2007). Thirty years of use and improvement of remote sensing, applied to epidemiology: from early promises to lasting frustration. Healh Place 13: 400–403.
- 21. Hamm NAS, Soares Magalhães RJ, Clements ACA (2015). Earth observation, spatial data quality, and neglected tropical diseases. PLoS Negl Trop Dis 9: e0004164. https://doi.org/10.1371/journal.pntd. 0004164 PMID: 26678393
- Walz Y, Wegmann M, Leutner B, Dech S, Vounatsou P, N'Goran EK, et al. (2015). Use of an ecologically relevant modelling approach to improve remote sensing-based schistosomiasis risk profiling. Geospat Health 10: 398. https://doi.org/10.4081/gh.2015.398 PMID: 26618326
- Grimes JET, Croll D, Harrison WE, Utzinger J, Freeman MC, Templeton MR (2014). The relationship between water, sanitation and schistosomiasis: a systematic review and meta-analysis. PLoS Negl Trop Dis 8: e3296. https://doi.org/10.1371/journal.pntd.0003296 PMID: 25474705
- 24. Grimes JET, Croll D, Harrison WE, Utzinger J, Freeman MC, Templeton MR, et al. (2015). The roles of water, sanitation and hygiene in reducing schistosomiasis: a review. Parasit Vectors 8: 156. https://doi.org/10.1186/s13071-015-0766-9 PMID: 25884172
- **25.** Esrey SA, Potash JB, Roberts L, Shiff C (1991). Effects of improved water supply and sanitation on ascariasis, diarrhoea, dracunculiasis, hookworm infection, schistosomiasis, and trachoma. Bull World Health Organ 69: 609–621. PMID: <u>1835675</u>
- Abdel-Rahman MS, El-Bahy MM, Malone JB, Thompson RA, El Bahy NM (2001). Geographic information systems as a tool for control program management for schistosomiasis in Egypt. Acta Trop 79: 49– 57. PMID: <u>11378141</u>

- Yapi RB, Hürlimann E, Houngbedji CA, N'Dri PB, Silué KD, Soro G, et al. (2014). Infection and co-infection with helminths and *Plasmodium* among school children in Côte d'Ivoire: results from a national cross-sectional survey. PLoS Negl Trop Dis 8: e2913. <u>https://doi.org/10.1371/journal.pntd.0002913</u> PMID: 24901333
- Kosinski KC, Bosompem KM, Stadecker MJ, Wagner AD, Plummer J, Durant JL, et al. (2011). Diagnostic accuracy of urine filtration and dipstick tests for *Schistosoma haematobium* infection in a lightly infected population of Ghanaian schoolchildren. Acta Trop 118: 123–127. https://doi.org/10.1016/j. actatropica.2011.02.006 PMID: 21354093
- Kulinkina AV, Kosinski KC, Plummer JD, Durant JL, Bosompem KM, Adjei MN, et al. (2017). Indicators of improved water access in the context of schistosomiasis transmission in rural Eastern Region, Ghana. Sci Total Environ 579: 1745–1755. https://doi.org/10.1016/j.scitotenv.2016.11.140 PMID: 27939198
- 30. Onori E, McCullough FS, Rosei L (1963). Schistosomiasis in the Volta Region of Ghana. Ann Trop Med Parasitol 57: 59–70. PMID: 13940173
- Kosinski KC, Kulinkina AV, Abrah AFA, Adjei MN, Breen KM, Chaudhry HM, et al. (2016). A mixedmethods approach to understanding water use and water infrastructure in a schistosomiasis-endemic community: case study of Asamama, Ghana. BMC Pub Health 16: 322.
- Kulinkina AV, Plummer JD, Chui KKH, Kosinski KC, Adomako-Adjei T, Egorov AI, et al. (2017). Physicochemical parameters affecting the perception of borehole water quality in Ghana. Int J Hyg Environ Health 220: 990–997. https://doi.org/10.1016/j.ijheh.2017.05.008 PMID: 28592357
- 33. Kulinkina AV, Kosinski KC, Liss A, Adjei MN, Ayamgah GA, Webb P, et al. (2016). Piped water consumption in Ghana: a case study of temporal and spatial patterns of clean water demand relative to alternative water sources in rural small towns. Sci Total Environ 559: 291–301. https://doi.org/10.1016/j.scitotenv.2016.03.148 PMID: 27070382
- **34.** Dungan JL, Perry JN, Dale MRT, Legendre P, Citron-Pousty S, et al. (2002). A balanced view of scale in spatial statistical analysis. Ecography 25: 626–640.
- United Nations Development Group. Indicators for Monitoring the Millennium Development Goals [Internet]. 2003 [cited 2018 Apr 05]. http://www.undp.org/content/dam/aplaws/publication/en/publications/ poverty-reduction/poverty-website/indicators-for-monitoring-the-mdgs/Indicators_for_Monitoring_the_ MDGs.pdf
- **36.** Walz Y (2014). Remote sensing for disease risk profiling: a spatial analysis of schistosomiasis in West Africa. PhD Thesis, University of Würzburg.
- ESRI [Internet]. Filtering DEMs; c2017 [cited 2018 Apr 05]. http://desktop.arcgis.com/en/arcmap/latest/ extensions/production-mapping/filtering-dems.htm
- Esch T, Heldens W, Hirner A, Keil M, Marconcini M, Roth A, et al. (2017). Breaking new ground in mapping human settlements from space—The Global Urban Footprint. J Photogramm Remote Sens 134: 30–42.
- Huete A, Didan K, Miura T, Rodriguez EP, Gao X, Ferreira LG (2002). Overview of the raidometric and biophysical performance of the MODIS vegetation indices. Remote Sens Environ 83: 195–213.
- United States Geological Survey (2017). Product guide: Landsat 8 surface reflectance-derived spectral indices (version 3.6). https://landsat.usgs.gov/sites/default/files/documents/si_product_guide.pdf (accessed 2018-04-05)
- 41. Huete AR (1988). A soil-adjusted vegetation index (SAVI). Remote Sens Environ 25: 295–309.
- **42.** Richter R (1996). A spatially adaptive fast atmospheric correction algorithm. Int J Remote Sens 17: 1201–1214.
- **43.** Tucker CJ (1979). Red and photographic infrared linear combinations for monitoring vegetation. Remote Sens Environ 8: 127–150.
- Qi J, Chehbouni A, Huete AR, Kerr YH, Sorooshian S (1994). A modified soil adjusted vegetation index. Remote Sens Environ 48: 119–126.
- McFeeters SK (1996). The use of the normalized difference water index (NDWI) in the delineation of open water features. Int J Remote Sens 17: 1425–1432.
- Xu H (2006). Modification of normalised difference water index (NDWI) to enhance open water features in remotely sensed imagery. Int J Remote Sens 27(14): 3025–3033.
- **47.** Strahler AN (1957). Quantitative analysis of watershed geomorphology. Trans Am Geophys Union 38: 913–920.
- **48.** Naing NN (2000). Easy way to learn standardization: direct and indirect methods. Malays J Med Sci 7: 10–15. PMID: 22844209
- **49.** Kulinkina AV (2017). Community based methods for schistosomiasis prediction and sustainable control in Ghana. PhD Thesis, Tufts University.

- Kampichler C, Wieland R, Calmé S, Weissenberger H, Arriaga-Weiss S (2010). Classification in conservation biology: a comparison of five machine-learning methods. Ecol Inform 5: 441–450.
- **51.** Li J, Alvarez B, Siwabessy J, Tran M, Huang Z, Przeslawski R, et al. (2017). Application of random forest and generalised linear model and their hybrid methods with geostatistical techniques to count data: predicting sponge species richness. Environ Model Softw 97: 112–129.
- 52. Hastie T, Tibshirani R, Friedman J (2009). The elements of statistical learning: data mining, inference, and prediction. Springer.
- **53.** Grömping U (2009). Variable importance assessment in regression: linear regression versus random forest. Am Stat 63: 308–319.
- Appleton CC (1978). Review of literature on abiotic factors influencing the distribution and life cycles of bilharziasis intermediate host snails. Malacol Rev 11: 1–25.
- 55. Chu KY (1966). Host-parasite relationship of *Bulinus truncatus* and *Schistosoma haematobium* in Iran. Bull World Health Org 34: 131–133. PMID: 5295559
- 56. Campbell SJ, Savage GB, Gray DJ, Atkinson JAM, Soares Magalhães RJ, Nery SV, et al. (2014). Water, sanitation, and hygiene (WASH): a critical component for sustainable soil-transmitted helminth and schistosomiasis control. PLoS Negl Trop Dis 8: e2651. https://doi.org/10.1371/journal.pntd. 0002651 PMID: 24722335